\begin{table}%T4
%\centering
\small
   \caption{SIPS objects which share common proper motions with
   objects found in other studies which are not themselves SIPS objects. These pairings were found during the cross referencing process and should not be regarded as a complete list. PA = Position Angle. For LEHPM objects see
   Pokorny (\cite{Pokorny03}), for NLTT objects see NLTT, $^1$ see
   Lasker et~al. (\cite{Lasker90}), $^2$ see Gizis et~al. (\cite{Gizis00}) and $^3$ see Schonfeld (\cite{Schonfeld86}). All pairs marked * were identified as common proper motion pairs by Luyten (\cite{Luyten88}).}
         \label{NLTTCPMObs}
\footnotesize
\begin{tabular}{lccrcl}
\hline\hline
Name&Position&$\mu$&PA&$\sigma_\mu$&Other\\
&&($''$/yr)&($^\circ$)&($''$/yr)&Name\\
\hline
\object{SIPS0551-8116}&05 51 54.60 $-$81 16 09.6&0.233&20&0.009&\\
\object{NLTT 15903}&05 52 38.16 $-$81 16 03.0&0.223&18&0.016&\\
\hline
\object{SIPS2126-7337}&21 26 20.62 $-$73 37 10.1&0.176&83&0.010&\\
\object{GSC 09334-00112}$^1$&21 26 30.94 $-$73 38 23.2&0.184&85&0.016&\\
\hline
\object{SIPS1954-6117}&19 54 51.61 $-$61 17 05.5&0.214&163&0.033&\\
\object{NLTT 48361}&19 54 49.14 $-$61 19 19.1&0.190&177&-&\\
\hline
\object{SIPS0447-5823}&04 47 11.69 $-$58 23 21.0&0.279&30.263&0.029&\\
\object{LEHPM 3838}& 04 47 13.38 $-$58 23 20.6&0.265&31.4&0.080&WT 155\\
\hline
\object{SIPS1858-4513}&18 58 21.39 $-$45 13 33.8&0.231&124&0.014&\\
\object{NLTT 47218}&18 58 16.68 $-$45 14 12.6&0.254&120&0.028&\\
\hline
\object{SIPS0333-4324}&03 33 18.57 $-$43 24 56.9&0.316&59&0.025&\\
\object{NLTT 11245}&03 33 18.03 $-$43 25 12.0&0.281&58&0.018&\\
\hline
\object{SIPS0123-3507}&01 23 24.56 $-$35 07 18.4&0.213&220&0.013&\object{LEHPM 1491}\\
\object{NLTT 4642}&01 23 29.93 $-$35 08 27.7&0.223&220&0.014&\object{LEHPM 1492}\\
\hline
\object{SIPS0116-3342}&01 16 31.19 $-$33 42 51.3&0.189&42&0.017&\object{LDS 3257b}*\\
\object{NLTT 4263}&01 16 32 78 $-$33 42 54.8&0.185&44&0.014&*\\
\hline
\object{SIPS0933-2752}&09 33 37.63 $-$27 52 47.4&0.328&297&0.022&\\
\object{NLTT 22073}&09 33 36.22 $-$27 52 25.0&0.311&289&0.017&\\
\hline
\object{SIPS1917-2748}&19 17 18.10 $-$27 48 54.0&0.209&133&0.028&\object{LDS 4807b}*\\
\object{NLTT 47615}&19 17 22.63 $-$27 47 32.26&0.195&141&0.008&*\\
\hline
\object{SIPS0132-2744}&01 32 35.91 $-$27 44 37.1&0.195&128&0.013&\object{NLTT 5135}*\\
\object{NLTT 5136}&01 32 37.48 $-$27 44 25.4&0.191&129&0.016&*\\
\hline
\object{SIPS0028-2651}&00 28 10.24 $-$26 51 24.7&0.189&208&0.015&\object{NLTT 1492}*\\
\object{NLTT 1491}&00 28 08.06 $-$26 52 25.8&0.198&207&0.021&*\\
\hline
\object{SIPS2147-2644}&21 47 44.64 $-$26 44 05.4&0.235&215&0.022&\object{2MASSW J2147446-264406}$^2$\\
\object{NLTT 52094}&21 47 47.00 $-$26 42 52.7&0.252&211&0.012&\\
\hline
\object{SIPS0227-2630}&02 27 25.50 $-$26 30 07.8&0.165&56&0.012&\object{NLTT 8057}*\\
\object{NLTT 8059}&02 27 24.37 $-$26 29 42.1&0.159&60&0.022&*\\
\hline
\object{SIPS1135-2017}&11 35 05.43 $-$20 17 26.3&0.245&279&0.017&\object{NLTT 27883}*\\
\object{NLTT 27884}&11 35 05.03 $-$20 16 57.6&0.231&281&0.024&*\\
\hline
\object{SIPS0133-1948}&01 33 08.77 $-$19 48 34.3&0.307&65&0.021&\object{NLTT 5169}*\\
\object{NLTT 5163}&01 33 05.55 $-$19 50 22.7&0.322&67&0.022&*\\
\hline
\object{SIPS1301-1848}&13 01 51.64 $-$18 48 40.5&0.335&286&0.016&\object{NLTT 32629}*\\
\object{NLTT 32636}&13 01 42.64 $-$18 47 23.2&0.321&287&0.021&*\\
\hline
\object{SIPS0532-1605}&05 32 33.82 $-$16 05 53.2&0.272&186&0.023&\\
\object{NLTT 15268}&05 32 42.06 $-$16 06 00.3&0.289&189&0.025&\\
\hline
\object{SIPS0536-1302}&05 36 07.79 $-$13 02 09.4&0.199&162&0.013&\object{NLTT 15357}*\\
\object{NLTT 15358}&05 36 08.58 $-$13 02 40.1&0.202&174&0.025&*\\
\hline
\object{SIPS2016-1100}&20 16 49.26 $-$11 00 13.7&0.349&206&0.019&\object{NLTT 48967}*\\
\object{NLTT 48973}&20 16 55.72 $-$10 58 54.6&0.318&206&-&*\\
\hline
\object{SIPS0321-0807}&03 21 46.26 $-$08 07 13.7&0.118&72&0.014&\\
\object{BD-08 638}$^3$&03 21 48.90 $-$08 06 10.58&0.127&76&0.019&\\
\hline
\object{SIPS1402-0447}&14 02 14.40 $-$04 47 53.2&0.231&254&0.018&\object{NLTT 36042}*\\
\object{NLTT 36043}&14 02 14.90 $-$04 48 09.0&0.214&250&0.021&*\\
\hline
\object{SIPS0023-0342}&00 23 30.92 $-$03 42 20.4&0.189&52.79&0.028&\object{NLTT 1210}*\\
\object{NLTT 1213/1214}&00 23 32.29 $-$03 42 28.4&0.232&55&-&*\\
\hline
\end{tabular}
\end{table}